Author's response to reviews

Title: Time course study of oxidative and nitrosative stress and antioxidant enzymes in K2Cr2O7-induced nephrotoxicity

Authors:

Jose Pedraza-Chaverri (pedraza@servidor.unam.mx)
Diana Barrrea (dianabarrera@hotmail.com)
Omar N Medina-Campos (mconoel@servidor.unam.mx)
Raymundo C Carbajal (clementecarbajal@hotmail.com)
Rogelio Hernandez-Pando (rhdezpando@hotmail.com)
Norma A Macias-Ruvalcaba (normaamr@hotmail.com)
Perla D Maldonado (m_cdeyanira@yahoo.com)
Marcos I Salcedo (ni_be_pb@hotmail.com)
Edilia Tapia (ediliatapia@htomail.com)
Liliana Saldivar (saldivar@servidor.unam.mx)
Maria E Castilla (elenacm@servidor.unam.mx)
Maria E Ibarra-Rubio (meir@servidor.unam.mx)

Version: 2 Date: 1 November 2004

Author's response to reviews:

1. References not included in the text were eliminated from the list of references.
2. Some syntaxes errors were corrected